f



Geophysical Research Letters^{*}

RESEARCH LETTER

10.1029/2022GL100391

Key Points:

- δ¹⁸O records from the Gulf of Mexico refine the chronology of Laurentide Ice Sheet advance and retreat through the Great Lakes
- Counterintuitively, from 17.6 to 11.3 ka, ice advanced during times of warming and retreated during times of cooling
- A feedback between meltwater, Atlantic Meridional Overturning Circulation strength, and temperature—paced by ice-sheet response time—may have set this anticorrelation

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

A. D. Wickert, awickert@umn.edu

Citation:

Wickert, A. D., Williams, C., Gregoire,
L. J., Callaghan, K. L., Ivanović,
R. F., Valdes, P. J., et al. (2023).
Marine-calibrated chronology of southern Laurentide Ice Sheet advance and retreat:
~2,000-year cycles paced by meltwater– climate feedback. *Geophysical Research Letters*, 50, e2022GL100391. https://doi. org/10.1029/2022GL100391

Received 17 JUL 2022 Accepted 29 DEC 2022 Corrected 1 AUG 2023

This article was corrected on 1 AUG 2023. See the end of the full text for details.

Author Contributions:

Conceptualization: Andrew D. Wickert, Lauren J. Gregoire, Ruža F. Ivanović, Lael Vetter, Carrie E. Jennings Data curation: Andrew D. Wickert, Carlie Williams

© 2023. The Authors. This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Marine-Calibrated Chronology of Southern Laurentide Ice Sheet Advance and Retreat: ~2,000-Year Cycles Paced by Meltwater–Climate Feedback

Andrew D. Wickert^{1,2,3}, Carlie Williams⁴, Lauren J. Gregoire⁵, Kerry L. Callaghan^{1,2,6,7}, Ruža F. Ivanović⁵, Paul J. Valdes⁸, Lael Vetter⁹, and Carrie E. Jennings^{2,10}

¹Saint Anthony Falls Laboratory, University of Minnesota, Minneapolis, MN, USA, ²Department of Earth & Environmental Sciences, University of Minnesota, Minneapolis, MN, USA, ³Deutsches GeoForschungsZentrum (GFZ): German Research Centre for Geosciences, Potsdam, Germany, ⁴College of Marine Sciences, University of South Florida, St. Petersburg, FL, USA, ⁵School of Earth and Environment, University of Leeds, Leeds, UK, ⁶Department of Earth and Environmental Sciences, Lamont-Doherty Earth Observatory, Columbia University, Palisades, NY, USA, ⁷Now at Department of Earth and Environmental Sciences, University of Illinois, Chicago, IL, USA, ⁸School of Geographical Sciences, University of Bristol, Bristol, UK, ⁹Department of Geosciences, University of Arizona, Tucson, AZ, USA, ¹⁰Freshwater Society, Saint Paul, MN, USA

Abstract Climatic warming following the Last Glacial Maximum caused the southern Laurentide Ice Sheet (LIS) to begin ~2,000-year cycles of retreat and readvance whose cause remains ambiguous. By developing a marine-calibrated chronology of southern LIS position, we counterintuitively demonstrate that between 17.6 and 11.3 ka, ice advanced during times of northern-hemisphere warming and retreated during times of northern-hemisphere cooling. Here we propose a cyclical feedback: Meltwater from ice retreat cooled the northern hemisphere by weakening the Atlantic Meridional Overturning Circulation (AMOC). This eventually lead to ice-sheet readvance, which reduced and rerouted meltwater discharge, and thereby allowed the AMOC to strengthen and the northern hemisphere to warm. Our data suggest that this antiphased ice-climate interaction, paced by ice-sheet response time, was initiated by synchronous warming and ice retreat ~18.7–17.6 ka (corresponding to the Erie "Interstade") and reached its apex during the Younger Dryas.

Plain Language Summary Twenty thousand years ago, a colder climate allowed glacial ice to flow from northern and eastern Canada into the Great Lakes region. But when climate began to warm, instead of simply melting from south to north, the margin of this ice front repeatedly retreated and re-advanced in 2,000-year cycles. First, we track these advances and retreats using records of ancient Gulf of Mexico water, stored in plankton shells retrieved from a sediment cores. Ice advance redirected water bearing "light" oxygen, with fewer neutrons, from Canada and the Great Lakes into the Mississippi River. When ice retreated, these waters found other courses to the sea, shifting Gulf of Mexico chemistry toward heavier oxygen (carrying more neutrons). Next, we ask why these ice-margin cycles occurred during a time of monotonically increasing solar radiation to the northern hemisphere. We propose that ice-sheet melt (during retreat) slowed Atlantic Ocean circulation, causing the northern hemisphere to cool and the ice sheet to gradually regrow. However, this same regrowth reduced meltwater flow to the ocean, causing the northern hemisphere to re-warm and eventually leading the ice margin to retreat once more. This cycle persisted from approximately 17,600 to 11,300 years ago, until North American ice became too small to sustain this feedback.

1. Introduction

Warming climate (Badgeley et al., 2020) associated with increasing northern-hemisphere summer insolation (Berger, 1978) triggered widespread ice retreat and the end of the global Last Glacial Maximum (LGM), approximately 19.5 ka (kiloannum: thousand calendar years before "present," BP, at 1950) (Clark et al., 2009). Laurentide Ice Sheet (LIS) meltwater produced ~63 m of global sea-level rise (Gowan et al., 2021)—approximately equal to the volume of all present-day ice masses combined—but the LIS did not smoothly retreat. Instead, the onset of deglaciation triggered repeated cycles of 150–400 km of ice retreat and readvance through the Great Lakes region (e.g., Licciardi et al., 1999; Mickelson & Colgan, 2003) (Figure 1: bases of black arrows).



Formal analysis: Andrew D. Wickert, Kerry L. Callaghan

Funding acquisition: Andrew D. Wickert Investigation: Andrew D. Wickert, Carlie Williams

Methodology: Andrew D. Wickert, Carlie Williams, Lauren J. Gregoire, Ruža F. Ivanović

Resources: Andrew D. Wickert, Carlie Williams, Paul J. Valdes Software: Andrew D. Wickert Validation: Andrew D. Wickert Visualization: Andrew D. Wickert Writing – original draft: Andrew D. Wickert

Writing – review & editing: Andrew D. Wickert, Lauren J. Gregoire, Kerry L. Callaghan, Ruža F. Ivanović, Paul J. Valdes, Lael Vetter, Carrie E. Jennings Both the driver of these cycles and their link to climate remain unclear. Chamberlin (1894) observed stacked till sequences, surmised that they resulted from repeated LIS retreat and readvance, and inferred that cooling produced advances and warming produced retreats (Flint, 1953; Taylor, 1897). However, later dating of glacial deposits (see summaries by Licciardi et al., 1999; Larson, 2011; Dalton et al., 2020), analysis of marine sediment cores (Emiliani, 1966), and applications of Milankovitch (1930) theory demonstrated that these dated glacial advances did not match periods of cooling from deep-ocean isotope records of paleoclimate. Instead, they occurred during a time of smoothly increasing northern-hemisphere summer insolation from the LGM to the end of the Pleistocene (Berger, 1978). Clark (1994) inferred that these cycles could be non-climatic in origin and instead reflect ice dynamics associated with basal-sediment deformation.

The hypothesis that the southern-LIS-margin oscillations occur independently of climate has sweeping implications for our understanding of ice-sheet response to climate change. If it is correct, little to no relationship between ice-margin position and climate could exist over thousands of years, thereby limiting our ability to predict ice-sheet behavior until we better understand subglacial sliding and till deformation (Clark, 1994; Zoet & Iverson, 2020), which are complex and can be difficult to characterize (Hansen & Zoet, 2022; Iverson & Petersen, 2011). On the other hand, if we can partially or fully disprove this hypothesis, we may gain confidence in simpler associations between climate change and ice-sheet response.

Glacial geological deposits (Licciardi et al., 1999) provide a proximal record of ice advance and retreat, but using these to develop an ice-margin chronology requires work to overcome: (a) the paucity of in situ organic material for high-precision AMS ¹⁴C dating during the early stages of the last deglaciation (Dalton et al., 2020), (b) the possibility of sampling reworked organic material that predates the deposit, (c) many ages providing maximum or minimum limiting dates on peak advance or retreat rather than a continuous record (Licciardi et al., 1999), and (d) the potential for local and asynchronous ice-lobe advance and retreat (Clark, 1994). These factors hinder efforts to directly compare the near-field record of bulk ice-marginal dynamics with patterns of changing climate and ocean circulation (Leydet et al., 2018; Licciardi et al., 1999; Teller & Leverington, 2004; Wickert, 2016). This limitation inhibits more quantitative data-driven analysis of inferred climate–ice-sheet links, including the oft-cited meltwater trigger for the Younger Dryas (Broecker et al., 1989; Condron & Winsor, 2012; Flower et al., 2011; Keigwin et al., 2018; Leydet et al., 2018; Murton et al., 2010; Tarasov & Peltier, 2005) and impacts of ice-sheet melt and meltwater rerouting on the AMOC and northern-hemisphere climate earlier in the last deglaciation (Ivanović et al., 2017, 2018; Ridge, 1997).

2. Methods

To unite these terrestrial records (see Table 1) into a continuous and high-resolution chronology that describes southern-LIS-margin fluctuations in bulk, we take advantage of changes in freshwater routing produced by advance and retreat of the Great Lakes ice lobes (Figure 2). Ice advance rerouted northward and eastward flowing drainages—and their isotopically light waters (Vetter et al., 2017)—toward the Mississippi and Gulf of Mexico (GoM), whereas ice retreat reduced Mississippi River outflow by opening alternate routes for these waters to reach the ocean (Carlson et al., 2007; Larson & Kincare, 2009; Licciardi et al., 1999; Obbink et al., 2010; Wickert, 2016). The combination of drainage rerouting and ice-sheet melt temporarily increased Mississippi River outflow to two to six times its ~2 × 10⁵ m³ s⁻¹ modern value (Wickert, 2016), and therefore significantly impacted the light-isotope inventory of the GoM (e.g., Kennett & Shackleton, 1975; Leventer et al., 1982).

Foraminiferal shells continuously record oxygen-isotope ratios (δ^{18} O) in the GoM and accumulate within seafloor sediments (e.g., Flower et al., 2004). When combined with information on seawater temperature (from Mg/Ca ratios also measured in these foraminifera; see Anand et al., 2003; Khider et al., 2015) and sea level (Schrag et al., 2002; Spratt & Lisiecki, 2016), these provide a record of past seawater oxygen-isotope ratios, which we term δ^{18} O_{ive-sw} (ice-volume-corrected seawater). Here we integrate these records with paleogeographic and paleoglaciological data from the duration of North American deglaciation to build a continuous, stratigraphically consistent, and extensively dated generalized ice-margin chronology (Table 1).

In order to use a GoM $\delta^{18}O_{ivc-sw}$ record as an indicator of ice-margin position, we must first ensure that the oxygen-isotope inventory of the GoM can unambiguously record ice-margin oscillations. This is complicated by the fact that the isotopic record is influenced both by changes in drainage routing—our desired target—and by changes in ice-sheet melt rate, which varied in space and time. To resolve the primary driver GoM





Figure 1. Deglacial North America, meltwater routing, and paleorecord locations. 15.5-ka ice extent (Dalton et al., 2020) and meltwater pathways (black arrows). Ice advance routed more water via the Mississippi toward the GoM, whereas ice retreat opened eastern outlets and/or northern meltwater outlets. Points indicate ocean core locations (LoDico et al., 2006; McManus et al., 2004; Williams, 2014; Williams et al., 2012), and Greenland is noted for its data-assimilated temperature record (Badgeley et al., 2020). The AMOC follows the red (warm) and blue (cold) arrows.

 $\delta^{18}O_{ivc-sw}$, we combine reconstructed drainage-basin extents (Wickert, 2014) with information on the $\delta^{18}O$ of meteoric water (Aharon, 2006), glacial melt (Ferguson & Jasechko, 2015), and GoM seawater (Wagner & Slowey, 2011). Based on these synthetic tests (Figure S3, Text S1.4, Table S1 in Supporting Information S1), we find that ice-sheet-driven changes in drainage-basin extent should dominate the variability in low- $\delta^{18}O$ water delivery to the GoM during all times but the Bølling–Allerød warm period, associated with Meltwater Pulse 1A (Wickert et al., 2013).

Following this synthetic test, we assembled a $\delta^{18}O_{ivc-sw}$ record using G. ruber (white) in core MD02-2550 (Figures 1 and 3a) (LoDico et al., 2006; Wickert, 2023; Williams, 2014; Williams et al., 2012). 65 ¹⁴C ages from the LGM to the early Holocene (Williams et al., 2010) underpin our age model, constructed using Bacon v2.5.1 (Blaauw & Christen, 2011) with the Marine20 radiocarbon calibration curve (Heaton et al., 2020): see Figure S1 in Supporting Information S1. This core has the most complete and well dated δ^{18} O chronology from the Gulf of Mexico across the late Pleistocene and early Holocene. Its finely laminated sediments indicate a lack of bioturbation that results from its anoxic basin origin (Meckler et al., 2008). It is located ~300 km south of the Mississippi River mouth (26.9462°N, 91.3457°W), and this distal position-seaward of the LGM shoreline-permits it to record the mixing of Mississippi-sourced waters with those from the Gulf of Mexico while decreasing its sensitivity to deltaic lobe switching by the Mississippi. We connected 487 measurements of δ^{18} O between 20.5 and 9.5 ka to this age model, alongside 295 Mg/Ca measurements as a proxy for sea-surface temperature (LoDico et al., 2006; Williams, 2014; Williams et al., 2012). Together, these provide a near-continuous record of $\delta^{18}O_{ivc-sw}$, from which we computed a 50-year moving average (using the software library by Wickert, 2019). Periods of low and high $\delta^{18}O_{ivc-sw}$ in this record closely match available terrestrial age control on the timing of ice-sheet advance and retreat, respectively (Table 1 and Text S1 in Supporting Information S1).

3. Results

Multiple methods converge on a ~2,000-year period of ice-margin fluctuations. Manually picking peaks in the GoM $\delta^{18}O_{ivc-sw}$ record (Figure 3a) yields a

 2.1 ± 0.6 kyr oscillation period. The Lomb–Scargle periodogram (Lomb, 1976; Scargle, 1982; VanderPlas, 2018) of these data displays a 1.63 kyr peak (Figure S2 in Supporting Information S1). Terrestrial data from glacial deposits constrain a 2.2 ± 0.7 kyr ice advance–retreat period (see references in Table 1 as well as Willman & Frye, 1970; Dreimanis & Karrow, 1972; Johnson et al., 1997; Karrow et al., 2000; Larson & Kincare, 2009; Larson, 2011).

We then binned the $\delta^{18}O_{ivc-sw}$ proxy southern LIS chronology into 50-year intervals, and correlated this against 50-year-mean spatially averaged temperatures from Greenland (Badgeley et al., 2020). At 100-year increments aligned with the start of each century BP, we computed correlation coefficients across a 950-year moving window (Figure 3c; Text S1.5 in Supporting Information S1), just approaching the half-period of the 2-kyr advance–retreat cycles and therefore approximately maximizing the number of data points that can be meaningfully correlated.

Prior to 17.6 ka, these data generally indicate a slight positive correlation between (approximately constant) Greenland temperature (Badgeley et al., 2020) and ice-margin change. After 11.3 ka, ice retreat and temperature increase more strongly align, consistent with modeling work demonstrating that Holocene-age LIS mass loss occurred dominantly due to surface ablation (Carlson et al., 2009). Between these times and during the main period of ice-margin oscillations, 17.6–11.3 ka, these data agree with neither the early-hypothesized direct correlation between temperature and ice-sheet extent (Flint, 1953; Taylor, 1897) nor the later-hypothesized

Table 1

Terrestrial and Marine Chronologies of Southern LIS Advance and Retreat

Event	Start date (δ ¹⁸ Ο, GoM) [ka]	Start date (terrestrial) [ka]	Terrestrial age source
Last Glacial Maximum (Nissouri Advance)	-	26.0	Muller and Calkin (1993); Clark et al. (2009)
Erie Retreat	18.7	~19.2	Mörner and Dreimanis (1973); Monaghan (1990); Muller and Calkin (1993); Ridge (1997)
Port Bruce Advance	17.6	~18.2	Fullerton (1980); Mickelson et al. (1983); Hansel and Johnson (1992); Licciardi et al. (1999); Dyke (2004)
Mackinaw Retreat	16.2	~16.1	C. F. M. Lewis et al. (1994); Karrow et al. (2000)
Port Huron Advance	15.2	15.6	Maher and Mickelson (1996)
Two Creeks Retreat	14.0	14.1	Kaiser (1994); Larson et al. (1994)
Onaway (or Greatlakean or Two Rivers) Advance	13.7	13.5	Broecker and Farrand (1963); Hansel and Mickelson (1988); Ridge (1997); Karrow et al. (2000); Rech et al. (2012)
Younger Dryas (Gribben) Retreat	12.8	12.9	Karrow et al. (2000); Toney et al. (2003); Richard and Occhietti (2005); Stanford (2010); Leydet et al. (2018)
Marquette Advance	11.7	11.6	Lowell et al. (1999)
Abitibi Retreat	10.5	10.6	Karrow et al. (2000); Fisher (2003); Breckenridge (2007); Derouin et al. (2007)

Note. The ~ symbol indicates poor age control: The Erie Retreat and Port Bruce Advance are constrained by maximum and minimum limiting dates, and direct dates of the Mackinaw Retreat are from a radiocarbon calibration-curve plateau (Heaton et al., 2020). Terrestrial dates are from deposits associated with the southern LIS. Marine dates are interpreted from the rise and fall of $\delta^{18}O_{ivc-sw}$ from the Gulf of Mexico record from core MD02–2550 (Brown, 2011; LoDico et al., 2006; Meckler et al., 2008; Williams, 2014; Williams et al., 2010, 2012).

non-correlation (Clark, 1994). Rather, the bulk of the data indicate a dominant anticorrelation: the southern LIS counterintuitively advanced during times of warming and retreated during times of cooling.

One mechanism by which LIS retreat and associated melt and northerly drainage rerouting could correlate with a cooling climate is through meltwater delivery to the North Atlantic and/or Arctic Oceans, \gtrsim 50°N (Smith & Gregory, 2009): see Text S2 in Supporting Information S1. This meltwater can form a low-salinity cap atop the seawater, increasing its density stratification and either inhibiting the AMOC directly or preconditioning the North Atlantic to switch AMOC modes. Weakening the AMOC decreases heat transport from the equator to higher latitudes (Ivanović et al., 2017, 2018).

To test an AMOC-driven relationship between LIS retreat and temperature, we correlated Pa/Th ratios from deepsea sediment (McManus et al., 2004; Ng et al., 2018)—as a residence-time-based proxy for AMOC strength against $\delta^{18}O_{ivc-sw}$. Because of the lower Pa/Th data density (one point per 86 ± 77 years) than the 50-year-increment temperature reanalysis product (Badgeley et al., 2020), we calculated 100-year moving averages of Pa/Th. We sampled this 100-year moving-average time series at the same 50-year increments as the temperature reanalysis and 50-year moving-average $\delta^{18}O_{ivc-sw}$ data, and then correlated Pa/Th ratios against $\delta^{18}O_{ivc-sw}$. Between 17.6 and 11.3 ka, AMOC strength generally anticorrelates with ice retreat, in line with the temperature records (Figure 3c). However, during the Bølling–Allerød (14.6–12.9 ka), positive correlations between AMOC strength and $\delta^{18}O_{ivc-sw}$ result from significant direct ice-sheet melt of low $\delta^{18}O$ waters (Vetter et al., 2017). These add a directly climate-driven signal to the Gulf of Mexico $\delta^{18}O_{ivc-sw}$ record that can counteract or overwhelm the influence of ice-sheet-extent (Figure S3 in Supporting Information S1). Prior to 17.6 ka, air temperature in Greenland remained approximately constant (Figure 3b), consistent with records from the Bahamas (Arienzo et al., 2015). In contrast, the 18.7 ka onset of the Erie Retreat corresponds to the beginning of AMOC fluctuations and weakening (Figure 3b) as well as to a drop in sea-surface temperature near Iberia (Naughton et al., 2016).

4. Discussion

Assuming that the Greenland temperature record is correlative with that of the Laurentide, as they lay adjacent to one another and no similar Laurentide temperature record exists, we propose the following sequence of





Figure 2. Southern LIS (white area) chronology and associated Mississippi River drainage-basin extent (heavy black line). Lakes are medium gray, land is dark gray, and oceans are light gray. Medium-gray lines indicate current (2022) political boundaries. Ice-sheet extents are modified slightly from their published dates (Dyke, 2004; Dyke et al., 2003) to fit the new marine chronology, for which ages on each panel are provided. Mississippi drainage-basin extents follow Wickert (2014), who used contours of ice-sheet extent through time (Dyke, 2004; Dyke et al., 2003) as approximate contours of ice-sheet thickness. The dashed line for the Marquette Advance indicates that Lake Agassiz drained into the Mississippi for at least some of this time (Fisher, 2003), though the exact dates remain unclear (Fisher & Breckenridge, 2022). The 8.8-ka age for the end of the Abitibi Retreat, marked by the Cochrane Advance, has an estimated error of ± 0.2 kyr (Breckenridge et al., 2012).

events (Figure 4): First, warming temperatures and increasing northern-hemisphere summer insolation (Ullman et al., 2015) caused the LIS to melt during the end of the LGM, resulting in the Erie Retreat ("Erie Inter-stade": Mörner & Dreimanis, 1973; Monaghan, 1990; Muller & Calkin, 1993; Ridge, 1997). This, combined



Geophysical Research Letters



Figure 3.





Figure 4. LIS–AMOC feedback loop. Ice-sheet retreat sends meltwater to the North Atlantic and/or Arctic Oceans, weakening the AMOC and causing northern-hemisphere cooling. This eventually leads to ice-sheet readvance, reducing meltwater discharge and allowing the AMOC to recover. This restores heat transport to the North Atlantic, leading to renewed ice-sheet retreat. This cycle was initiated by increased solar radiation at the end of the LGM and terminated in the early Holocene following significant ice-sheet mass loss and retreat.

with the opening of more northerly meltwater outlets (capable of weakening the AMOC: Smith & Gregory, 2009), initiated a cycle in which (a) the LIS retreated and sent meltwater to the North Atlantic and/or Arctic Oceans; (b) this meltwater capped the North Atlantic, slowing the AMOC and reducing northern-hemisphere temperatures; (c) this cooling induced southern LIS readvance and associated water storage in newly-produced glacial ice; and (d) as freshwater delivery to the oceans decreased, the AMOC recovered and northern-hemisphere temperatures rose. This then induced renewed ice-sheet retreat, continuing the cycle. Final termination of this cycle occurred in the early Holocene when peak northern-hemisphere summer insolation (Berger, 1978; Carlson et al., 2008), an ice-free Hudson Bay (Ullman et al., 2016), and a diminished Laurentide Ice Sheet tipped the scales toward a dominance of climatic forcing over ice-sheet response.

To produce the observed anticorrelation between ice-margin migration and both temperature and AMOC strength, AMOC response to meltwater input must be fast, whereas the time for ice-sheet response to changing temperature must be approximately half of the measured ~2 kyr advance–retreat cycle. Ocean-circulation modeling indicates that the AMOC responds to meltwater inputs within a decade (Condron & Winsor, 2012; Ivanović et al., 2017, 2018). A characteristic response time (Johannesson et al., 1989) of the southern LIS to a mass-balance perturbation is roughly 500–2000 years, based on the thickness of the ice lobes and their proximal up-ice contributing areas (1–2 km: Margold et al., 2015; Margold et al., 2015; Gowan et al., 2021) and characteristic southern LIS ablation rates (1–2 m yr⁻¹: Peltier et al., 2015; Ivanović et al., 2017; Ivanović et al., 2018). The large scale gap between these response times supports the hypothesized close link between ocean circulation and temperature, with ~2,000-year cycles paced by a ~1000-year ice-sheet response time (Figure 4).

The proposed feedback may explain periods of rapid climate and ice-sheet change and provide targets for improved models of climate dynamics. Persistent ice retreat and northerly routed melt during the Younger Dryas correlates strongly with northern-hemisphere cooling (Condron & Winsor, 2012; Ivanović et al., 2017; Keigwin et al., 2018; Meissner & Clark, 2006; Murton et al., 2010; Smith & Gregory, 2009). Likewise, southward meltwater rerouting and ice-sheet storage during the Port Huron Advance (15.4–14.0 ka) could strengthen the AMOC (Smith & Gregory, 2009) during the Břlling warming, though the plausibility of this mechanism modifying ocean-scale climate also depends on the magnitude of Arctic-directed melt, for example, from the Cordilleran–Laurentide ice-saddle collapse (Gregoire et al., 2012; Ivanović et al., 2017), which would weaken the AMOC (Smith & Gregory, 2009). Importantly, climate-model response to freshwater forcing varies based on model structure and boundary conditions (e.g., comparing Ivanović et al., 2018; Obase et al., 2021; Romé et al., 2022; Sun et al., 2022), and models may be over-sensitive (as suggested by He & Clark, 2022) or under-sensitive (Valdes, 2011) to freshwater forcing. By presenting time-series data on freshwater routing to the oceans and their correlations with atmospheric temperature and ocean circulation, we provide ingredients to test and help improve such dynamical models.

Although the LIS was the largest northern-hemisphere ice sheet, the Greenland and Fennoscandian ice sheets may also have influenced—and been influenced by—ocean circulation. The Greenland Ice Sheet has a response time similar to that of the southern LIS (Cuffey & Marshall, 2000), and it also retreated and

Figure 3. Paleorecords and correlations. The gray field highlights the time of anticorrelation between temperature and ice extent, and the diagonally hatched region denotes the Bølling–Allerød, during which we expect $\delta^{18}O_{ivc-sw}$ to be significantly influenced by transient ice melt and therefore no longer be a good indicator of ice-sheet extent. Lines provide 100-year moving averages. (a) $\delta^{18}O_{ivc-sw}$ record from core MD02–2550 in the Gulf of Mexico (LoDico et al., 2006; Williams, 2014; Williams et al., 2012). Error bars indicate 2- σ uncertainties in both age and $\delta^{18}O_{ivc-sw}$. Aside from the LGM, labeled periods of advance (bold blue-gray) and retreat (roman brown) denote their start times, with arrows indicating durations (Table 1). (b) Greenland temperature anomalies from present (Badgeley et al., 2020) and Atlantic Pa/Th ratios as a proxy for AMOC strength (higher Pa/Th: weaker AMOC). (c) Correlations between $\delta^{18}O_{ivc-sw}$, indicating ice-sheet extent, and both Greenland temperature and North Atlantic Pa/Th ratios. Markers are plotted for correlations within each 950-year window. Marker sizes are proportional to one minus Pearson's *p* (with no correlation as the null hypothesis), and filled markers indicate that the correlations are statistically significant to the *p* < 0.05 level.

19448007, 2023, 10, Downle

10.1029/2022GL100391 by

GFZ,

Wiley Online Library

on [09/08/2023]

thinned during the Younger Dryas (Carlson et al., 2021; Rainsley et al., 2018). Geological evidence (Boswell et al., 2019; Toucanne et al., 2009, 2015) alongside model results (Ivanović et al., 2018) from both the most recent and prior glacial periods indicate that Fennoscandian Ice Sheet melt likewise corresponds to periods of northern-hemisphere cooling and AMOC slowdown. Moving beyond our LIS focus-itself a product of the utility of the GoM records—and considering northern-hemisphere ice-margin fluctuations in sum could improve our understanding of climate-meltwater interactions and plausibly bolster our observed climateice-margin anticorrelation.

5. Conclusions

The proposed ice-ocean coupling presents a mechanism for an ice sheet to temporarily prolong glacial-stage climatic conditions by creating a protective barrier of freshwater that inhibits heat transport from the tropics. It explains the enigmatic observation that the southern LIS retreated during the Younger Dryas cold period (Dalton et al., 2020) and extends this interpretation of antiphased climate change and ice-sheet response back to 17.6 ka, though with a weaker feedback prior to the opening of major northerly meltwater outlets (Carlson et al., 2007; Condron & Winsor, 2012; Keigwin et al., 2018; Murton et al., 2010; Smith & Gregory, 2009). Together, these data indicate that the LIS may have controlled North Atlantic climate cyclicity across the majority of the Late Pleistocene deglaciation. Over the coming century, we may observe whether retreat of the much-smaller Greenland Ice Sheet produces a similar AMOC feedback (Bakker et al., 2016; Ivanović et al., 2018) and/or whether Antarctic Ice Sheet meltwater and its dynamics follow modeled predictions to a different end (Bronselaer et al., 2018; Fogwill et al., 2015) by enhancing deepwater temperatures, thermally eroding the ice-sheet base, and enabling a destabilizing positive feedback toward further mass loss.

Data Availability Statement

Both raw and processed data are available from Wickert (2023). In addition, these analyses are based on the "d18O-ivc-sw" library, available on GitHub (https://github.com/awickert/d18O-ivc-sw) and archived on Zenodo (Wickert, 2019).

References

Aharon, P. (2006). Entrainment of meltwaters in hyperpycnal flows during deglaciation superfloods in the Gulf of Mexico. Earth and Planetary Science Letters, 241(1-2), 260-270, https://doi.org/10.1016/j.eps1.2005.10.034

- Anand, P., Elderfield, H., & Conte, M. H. (2003). Calibration of Mg/Ca thermometry in planktonic foraminifera from a sediment trap time series. Paleoceanography, 18(2), 1050. https://doi.org/10.1029/2002PA000846
- Arienzo, M. M., Swart, P. K., Pourmand, A., Broad, K., Clement, A. C., Murphy, L. N., et al. (2015). Bahamian speleothem reveals temperature decrease associated with Heinrich stadials. Earth and Planetary Science Letters, 430, 377-386. https://doi.org/10.1016/j.epsl.2015.08.035

Badgeley, J. A., Steig, E. J., Hakim, G. J., & Fudge, T. J. (2020). Greenland temperature and precipitation over the last 20 000 years using data assimilation. Climate of the Past, 16(4), 1325-1346. https://doi.org/10.5194/cp-16-1325-2020

Bakker, P., Schmittner, A., Lenaerts, J. T. M., Abe-Ouchi, A., Bi, D., van den Broeke, M. R., et al. (2016). Fate of the Atlantic meridional overturning circulation: Strong decline under continued warming and Greenland melting. Geophysical Research Letters, 43(23), 12252–12260. https://doi.org/10.1002/2016GL070457

Berger, A. L. (1978). Long-term variations of caloric insolation resulting from the Earth's orbital elements. Quaternary Research, 9(02), 139–167. https://doi.org/10.1016/0033-5894(78)90064-9

Blaauw, M., & Christen, J. A. (2011). Flexible paleoclimate age-depth models using an autoregressive gamma process. Bayesian Analysis, 6(3), 457-474. https://doi.org/10.1214/11-BA618

Boswell, S. M., Toucanne, S., Pitel-Roudaut, M., Creyts, T. T., Eynaud, F., & Bayon, G. (2019). Enhanced surface melting of the fennoscandian ice sheet during periods of north Atlantic cooling. Geology, 47(7), 664-668. https://doi.org/10.1130/G46370.1

Breckenridge, A. (2007). The lake superior varve stratigraphy and implications for eastern lake Agassiz outflow from 10,700 to 8900 cal ybp (9.5-8.0 ^{14}C ka). Palaeogeography, Palaeoclimatology, Palaeoecology, 246(1), 45-61. https://doi.org/10.1016/j.palaeo.2006.10.026

Breckenridge, A., Lowell, T. V., Stroup, J. S., & Evans, G. (2012). A review and analysis of varve thickness records from glacial lake Ojibway (Ontario and Quebec, Canada). Quaternary International, 260, 43-54. https://doi.org/10.1016/i.quaint.2011.09.031

Broecker, W. S., & Farrand, W. R. (1963). Radiocarbon age of the two creeks forest bed, Wisconsin. Bulletin of the Geological Society of America, 74, 795-802. https://doi.org/10.1130/0016-7606(1963)74[795:RAOTTC]2.0.CO;2

Broecker, W. S., Kennett, J. P., Flower, B. P., Teller, J. T., Trumbore, S., Bonani, G., & Wolfli, W. (1989). Routing of meltwater from the Laurentide Ice Sheet during the younger Dryas cold episode. Nature, 341(6240), 318-321. https://doi.org/10.1038/341318a0

Bronselaer, B., Winton, M., Griffies, S. M., Hurlin, W. J., Rodgers, K. B., Sergienko, O. V., et al. (2018). Change in future climate due to Antarctic meltwater. Nature, 564(7734), 53-58. https://doi.org/10.1038/s41586-018-0712-z

Brown, E. A. (2011). Initial ablation of the Laurentide Ice Sheet based on Gulf of Mexico sediments (Unpublished master's thesis). University of South Florida.

Carlson, A. E., Anslow, F. S., Obbink, E. A., LeGrande, A. N., Ullman, D. J., & Licciardi, J. M. (2009). Surface-melt driven Laurentide Ice Sheet retreat during the early holocene. Geophysical Research Letters, 36(24), L24502. https://doi.org/10.1029/2009GL040948

ADW was supported by the US Department of Defense through the National Defense Science & Engineering Graduate Fellowship Program, by the US National Science Foundation Graduate Research Fellowship under Grant DGE 1144083, by the McKnight Foundation through a McKnight Land-Grant Professorship awarded through the University of Minnesota, and by a Humboldt-Forschungsstipendium from the Alexander von Humboldt Foundation. Conversations with Andy Breckenridge helped to inform and shape this manuscript.

- Carlson, A. E., Clark, P. U., Haley, B. A., Klinkhammer, G. P., Simmons, K., Brook, E. J., & Meissner, K. J. (2007). Geochemical proxies of North American freshwater routing during the younger Dryas cold event. Proceedings of the National Academy of Sciences of the United States of America, 104(16), 6556–6561. https://doi.org/10.1073/pnas.0611313104
- Carlson, A. E., LeGrande, A. N., Oppo, D. W., Came, R. E., Schmidt, G. A., Anslow, F. S., et al. (2008). Rapid early Holocene deglaciation of the Laurentide Ice Sheet. *Nature Geoscience*, 1(9), 620–624. https://doi.org/10.1038/ngeo285
- Carlson, A. E., Reyes, A. V., Gusterson, E., Axford, Y., Wilcken, K. M., & Rood, D. H. (2021). Direct evidence for thinning and retreat of the southernmost Greenland ice sheet during the Younger Dryas. *Quaternary Science Reviews*, 267, 107105. https://doi.org/10.1016/j. quascirev.2021.107105
- Chamberlin, T. C. (1894). Glacial phenomena of North America. In J. Geikie (Ed.), *The great ice age and its relation to the antiquity of man* (pp. 724–752). Edward Stanford.
- Clark, P. U. (1994). Unstable behavior of the Laurentide Ice Sheet over deforming sediment and its implications for climate change. *Quaternary Research*, 41(01), 19–25. https://doi.org/10.1006/qres.1994.1002
- Clark, P. U., Dyke, A. S., Shakun, J. D., Carlson, A. E., Clark, J., Wohlfarth, B., et al. (2009). The last glacial maximum. Science, 325(5941), 710–714. https://doi.org/10.1126/science.1172873
- Condron, A., & Winsor, P. (2012). Meltwater routing and the younger Dryas. Proceedings of the National Academy of Sciences of the United States of America, 109(49), 19928–19933. https://doi.org/10.1073/pnas.1207381109
- Cuffey, K. M., & Marshall, S. J. (2000). Substantial contribution to sea-level rise during the last interglacial from the Greenland ice sheet. Nature, 404(6778), 591–594. https://doi.org/10.1038/35007053
- Dalton, A. S., Margold, M., Stokes, C. R., Tarasov, L., Dyke, A. S., Adams, R. S., et al. (2020). An updated radiocarbon-based ice margin chronology for the last deglaciation of the North American Ice Sheet Complex. *Quaternary Science Reviews*, 234, 106223. https://doi.org/10.1016/j. quascirev.2020.106223
- Derouin, S. A., Lowell, T. V., & Hajdas, I. (2007). Landscape evolution and deglaciation of the Upper Peninsula, Michigan: An examination of chronology and stratigraphy in kettle lake cores. *Journal of Great Lakes Research*, 33(4), 875–886. https://doi.org/10.3394/0380-1330(2007) 33[875:LEADOT]2.0.CO;2
- Dreimanis, A., & Karrow, P. F. (1972). Glacial history of the great lakes–St. Lawrence region, the classification of the Wisconsin(an) stage, and its correlatives. 24th International Geological Congress (pp. 5–15).
- Dyke, A. S. (2004). An outline of North American deglaciation with emphasis on central and northern Canada. In J. Ehlers & P. L. Gibbard (Eds.), Quaternary glaciations–extent and chronology—Part II: North America, Series title: Developments in quaternary sciences (Vol. 2, pp. 373–424). Elsevier Science. https://doi.org/10.1016/S1571-0866(04)80209-4
- Dyke, A. S., Moore, A., & Robertson, L. (2003). Deglaciation of North America. Ottawa: Natural Resources Canada.
- Emiliani, C. (1966). Paleotemperature analysis of Caribbean cores P6304-8 and P6304-9 and a generalized temperature curve for the past 425,000 years. *The Journal of Geology*, 74(2), 109–124. https://doi.org/10.1086/627150
- Ferguson, G., & Jasechko, S. (2015). The isotopic composition of the Laurentide Ice Sheet and fossil groundwater. *Geophysical Research Letters*, 42(12), 1–6. https://doi.org/10.1002/2015GL064106.Received
- Fisher, T. G. (2003). Chronology of glacial lake Agassiz meltwater routed to the Gulf of Mexico. *Quaternary Research*, 59(2), 271–276. https://doi.org/10.1016/s0033-5894(03)00011-5
- Fisher, T. G., & Breckenridge, A. (2022). Relative lake level reconstructions for glacial lake Agassiz spanning the Herman to Campbell levels. *Quaternary Science Reviews*, 294, 107760. https://doi.org/10.1016/j.quascirev.2022.107760
- Flint, R. F. (1953). Probable Wisconsin substages and late-Wisconsin events in northeastern United States and southeastern Canada. Bulletin of the Geological Society of America, 64(8), 897–920. https://doi.org/10.1130/0016-7606(1953)64[897:PWSALE]2.0.CO;2
- Flower, B. P., Hastings, D. W., Hill, H. W., & Quinn, T. M. (2004). Phasing of deglacial warming and Laurentide Ice Sheet meltwater in the Gulf of Mexico. Geology, 32(7), 597. https://doi.org/10.1130/G20604.1
- Flower, B. P., Williams, C., Hill, H. W., & Hastings, D. W. (2011). Laurentide Ice Sheet meltwater and the Atlantic meridional overturning circulation during the last glacial cycle: A view from the Gulf of Mexico. *Geophysical Monograph Series*, 193, 39–56. https://doi. org/10.1029/2010GM001016
- Fogwill, C. J., Phipps, S. J., Turney, C. S. M., & Golledge, N. R. (2015). Sensitivity of the Southern Ocean to enhanced regional Antarctic ice sheet meltwater input. *Earth's Future*, 3(10), 317–329. https://doi.org/10.1002/2015EF000306
- Fullerton, D. S. (1980). Preliminary correlation of post-Erie interstadial events (16k-10k radiocarbon years before present), Central and eastern great lakes region, and Hudson, Champlain, and St. Lawrence Lowlands, United States and Canada (Vol. 1089). U.S. Govt. Print. Off. (Series Title: Professional Paper). https://doi.org/10.3133/pp1089
- Gowan, E. J., Zhang, X., Khosravi, S., Rovere, A., Stocchi, P., Hughes, A. L. C., et al. (2021). A new global ice sheet reconstruction for the past 80 000 years. *Nature Communications*, 12(1), 1199. https://doi.org/10.1038/s41467-021-21469-w
- Gregoire, L. J., Payne, A. J., & Valdes, P. J. (2012). Deglacial rapid sea level rises caused by ice-sheet saddle collapses. *Nature*, 487(7406), 219–222. https://doi.org/10.1038/nature11257
- Hansel, A. K., & Johnson, W. H. (1992). Fluctuation of the lake Michigan lobe during the late Wisconsin subepisode. Sveriges Geologiska Undersokning, 81, 133–144.
- Hansel, A. K., & Mickelson, D. M. (1988). A reevaluation of the timing and causes of high lake phases in the Lake Michigan basin. *Quaternary Research*, 29(2), 113–128. https://doi.org/10.1016/0033-5894(88)90055-5
- Hansen, D. D., & Zoet, L. K. (2022). Characterizing sediment flux of deforming glacier beds. Journal of Geophysical Research: Earth Surface, 127(4), e2021JF006544. https://doi.org/10.1029/2021JF006544
- He, F., & Clark, P. U. (2022). Freshwater forcing of the Atlantic meridional overturning circulation revisited. Nature Climate Change, 12(5), 449–454. https://doi.org/10.1038/s41558-022-01328-2
- Heaton, T. J., Köhler, P., Butzin, M., Bard, E., Reimer, R. W., Austin, W. E., et al. (2020). Marine20 The marine radiocarbon age calibration curve (0-55,000 cal BP). *Radiocarbon*, 62(4), 779–820. https://doi.org/10.1017/RDC.2020.68
- Ivanović, R. F., Gregoire, L. J., Burke, A., Wickert, A. D., Valdes, P. J., Ng, H. C., et al. (2018). Acceleration of northern ice sheet melt induces AMOC slowdown and northern cooling in simulations of the early last deglaciation. *Paleoceanography and Paleoclimatology*, 33(7), 807–824. https://doi.org/10.1029/2017PA003308
- Ivanović, R. F., Gregoire, L. J., Wickert, A. D., Valdes, P. J., & Burke, A. (2017). Collapse of the North American ice saddle 14,500 years ago caused widespread cooling and reduced ocean overturning circulation. *Geophysical Research Letters*, 44(1), 383–392. https://doi. org/10.1002/2016GL071849
- Iverson, N. R., & Petersen, B. B. (2011). A new laboratory device for study of subglacial processes: First results on ice-bed separation during sliding. *Journal of Glaciology*, 57(206), 1135–1146. https://doi.org/10.3189/002214311798843458

- Johannesson, T., Raymond, C., & Waddington, E. (1989). Time-scale for adjustment of glaciers to changes in mass balance. *Journal of Glaciology*, 35(121), 355–369. https://doi.org/10.1017/S002214300000928X
- Johnson, W. H., Hansel, A. K., Bettis, E., Arthur, I. I. I., Karrow, P. F., Larson, G. J., et al. (1997). Late quaternary temporal and event classifications, Great Lakes region, North America. *Quaternary Research*, 47(1), 1–12. https://doi.org/10.1006/qres.1996.1870
- Kaiser, K. F. (1994). Two creeks interstade dated through dendrochronology and AMS. *Quaternary Research*, 42(3), 288–298. https://doi. org/10.1006/qres.1994.1079
- Karrow, P. F., Dreimanis, A., & Barnett, P. J. (2000). A proposed diachronic revision of late quaternary time-stratigraphic classification in the eastern and northern great lakes area. *Quaternary Research*, 54, 1–12. https://doi.org/10.1006/qres.2000.2144
- Keigwin, L. D., Klotsko, S., Zhao, N., Reilly, B., Giosan, L., & Driscoll, N. W. (2018). Deglacial floods in the Beaufort Sea preceded younger Dryas cooling. *Nature Geoscience*, 11(8), 599–604. https://doi.org/10.1038/s41561-018-0169-6
- Kennett, J. P., & Shackleton, N. J. (1975). Laurentide Ice Sheet meltwater recorded in Gulf of Mexico deep-sea cores. Science (New York, N.Y.), 188(4184), 147–150. https://doi.org/10.1126/science.188.4184.147
- Khider, D., Huerta, G., Jackson, C., Stott, L. D., & Emile-Geay, J. (2015). A Bayesian, multivariate calibration for *Globigerinoides ruber* Mg/Ca. *Geochemistry, Geophysics, Geosystems*, 16(9), 2916–2932. https://doi.org/10.1002/2015GC005844
- Larson, G. J. (2011). Ice-margin fluctuations at the end of the Wisconsin episode, Michigan, USA. In J. Ehlers, P. L. Gibbard, & P. D. Hughes (Eds.), Quaternary glaciations - Extent and chronology: A closer look (Vol. 15, pp. 489–497). Elsevier Inc. https://doi.org/10.1016/ B978-0-444-53447-7.00037-4
- Larson, G. J., & Kincare, K. (2009). Late quaternary history of the eastern mid-continent region, USA. In R. J. Schaetzl, J. Darden, & D. Brandt (Eds.), *Michigan geography and geology* (pp. 69–90). Custom Publishing.
- Larson, G. J., Lowell, T. V., & Ostrom, N. E. (1994). Evidence for the two creeks interstade in the lake Huron basin. Canadian Journal of Earth Sciences, 31(5), 793–797. https://doi.org/10.1139/e94-072
- Leventer, A., Williams, D. F. D. F., & Kennett, J. P. (1982). Dynamics of the Laurentide Ice Sheet during the last deglaciation: Evidence from the Gulf of Mexico. *Earth and Planetary Science Letters*, 59(1), 11–17. https://doi.org/10.1016/0012-821x(82)90112-1
- Lewis, C. F. M., Moore, T. C., Rea, D. K., Dettman, D. L., Smith, A. M., & Mayer, L. A. (1994). Lakes of the Huron basin: Their record of runoff from the Laurentide Ice Sheet. *Quaternary Science Reviews*, 13(9–10), 891–922. https://doi.org/10.1016/0277-3791(94)90008-6
- Leydet, D. J., Carlson, A. E., Teller, J. T., Breckenridge, A., Barth, A. M., Ullman, D. J., et al. (2018). Opening of glacial Lake Agassiz's eastern outlets by the start of the Younger Dryas cold period. *Geology*, 46(2), 155–158. https://doi.org/10.1130/G39501.1
- Licciardi, J. M., Teller, J. T., & Clark, P. U. (1999). Freshwater routing by the Laurentide Ice Sheet during the last deglaciation. In P. U. Clark, R. S. Webb, & L. D. Keigwin (Eds.), *Mechanisms of global climate change at millennial time scales* (Vol. 112, pp. 177–201). American Geophysical Union. https://doi.org/10.1029/GM112p0177
- LoDico, J. M., Flower, B. P., & Quinn, T. M. (2006). Subcentennial-scale climatic and hydrologic variability in the Gulf of Mexico during the early Holocene. *Paleoceanography*, 21(3), PA3015. https://doi.org/10.1029/2005PA001243
- Lomb, N. R. (1976). Least-squares frequency analysis of unequally spaced data. Astrophysics and Space Science, 39(2), 447–462. https://doi. org/10.1007/BF00648343
- Lowell, T. V., Larson, G. J., Hughes, J. D., & Denton, G. H. (1999). Age verification of the lake Gribben forest bed and the younger Dryas advance of the laurentide Ice Sheet. *Canadian Journal of Earth Sciences*, 36(3), 383–393. https://doi.org/10.1139/e98-095
- Maher, L. J., & Mickelson, D. M. (1996). Palynological and radiocarbon evidence for deglaciation events in the Green Bay Lobe, Wisconsin. *Quaternary Research*, 46(3), 251–259. https://doi.org/10.1006/qres.1996.0064
- Margold, M., Stokes, C. R., & Clark, C. D. (2015). Ice streams in the Laurentide Ice Sheet: Identification, characteristics and comparison to modern ice sheets. *Earth-Science Reviews*, 143, 117–146. https://doi.org/10.1016/j.earscirev.2015.01.011
- McManus, J. F., Francois, R., Gherardi, J.-M., Keigwin, L. D., & Brown-Leger, S. (2004). Collapse and rapid resumption of Atlantic meridional circulation linked to deglacial climate changes. *Nature*, 428(6985), 834–837. https://doi.org/10.1038/nature02494
- Meckler, A., Schubert, C., Hochuli, P., Plessen, B., Birgel, D., Flower, B., et al. (2008). Glacial to Holocene terrigenous organic matter input to sediments from Orca Basin, Gulf of Mexico — A combined optical and biomarker approach. *Earth and Planetary Science Letters*, 272(1–2), 251–263. https://doi.org/10.1016/j.epsl.2008.04.046
- Meissner, K. J., & Clark, P. U. (2006). Impact of floods versus routing events on the thermohaline circulation. Geophysical Research Letters, 33(15), L15704. https://doi.org/10.1029/2006GL026705
- Mickelson, D. M., Clayton, L., Fullerton, D. S., & Borns, H. W. (1983). The Late Wisconsin glacial record of the Laurentide Ice Sheet in the United States (Vol. 1).
- Mickelson, D. M., & Colgan, P. M. (2003). The southern laurentide Ice Sheet. In A. R. Gillespie, S. C. Porter, & B. F. Atwater (Eds.), The quaternary period in the United States (Vol. 1, pp. 1–16). https://doi.org/10.1016/S1571-0866(03)01001-7
- Milankovitch, M. (1930). Mathematische klimalehre und astronomische theorie der klimaschwankungen. In W. Köppen & R. Geiger (Eds.), Handbuch der Klimatologie (Vol. 1). Gebrüder Borntraeger.
- Monaghan, G. W. (1990). Systematic variation in the clay-mineral composition of till sheets: Evidence for the erie interstade in the lake Michigan basin. In A. F. Schneider & G. S. Fraser (Eds.), GSA special papers: Late quaternary history of the lake Michigan basin (Vol. 251, pp. 43–50). Geological Society of America. https://doi.org/10.1130/SPE251-p43
- Mörner, N.-A., & Dreimanis, A. (1973). The Erie interstade. Geological Society of America, 136(1958), 107–134. https://doi.org/10.1130/ MEM136-p107
- Muller, E. H., & Calkin, P. E. (1993). Timing of Pleistocene glacial events in New York state. Canadian Journal of Earth Sciences, 30(9), 1829–1845. https://doi.org/10.1139/e93-161
- Murton, J. B., Bateman, M. D., Dallimore, S. R., Teller, J. T., & Yang, Z. (2010). Identification of younger Dryas outburst flood path from Lake Agassiz to the Arctic ocean. *Nature*, 464(7289), 740–743. https://doi.org/10.1038/nature08954
- Naughton, F., Sanchez Goñi, M., Rodrigues, T., Salgueiro, E., Costas, S., Desprat, S., et al. (2016). Climate variability across the last deglaciation in NW Iberia and its margin. *Quaternary International*, 414, 9–22. https://doi.org/10.1016/j.quaint.2015.08.073
- Ng, H. C., Robinson, L. F., McManus, J. F., Mohamed, K. J., Jacobel, A. W., Ivanovic, R. F., et al. (2018). Coherent deglacial changes in western Atlantic Ocean circulation. *Nature Communications*, 9(1), 1–10. https://doi.org/10.1038/s41467-018-05312-3
- Obase, T., Abe-Ouchi, A., & Saito, F. (2021). Abrupt climate changes in the last two deglaciations simulated with different Northern ice sheet discharge and insolation. *Scientific Reports*, 11(1), 22359. https://doi.org/10.1038/s41598-021-01651-2
- Obbink, E. A., Carlson, A. E., & Klinkhammer, G. P. (2010). Eastern North American freshwater discharge during the Bolling-Allerod warm periods. *Geology*, 38(2), 171–174. https://doi.org/10.1130/G30389.1
- Peltier, W. R., Argus, D. F., & Drummond, R. (2015). Space geodesy constrains ice age terminal deglaciation: The global ICE-6G_c (VM5a) model. Journal of Geophysical Research: Solid Earth, 120(1), 450–487. https://doi.org/10.1002/2014JB011176

- Rainsley, E., Menviel, L., Fogwill, C. J., Turney, C. S., Hughes, A. L., & Rood, D. H. (2018). Greenland ice mass loss during the younger Dryas driven by Atlantic meridional overturning circulation feedbacks. *Scientific Reports*, 8(1), 11307. https://doi.org/10.1038/s41598-018-29226-8
 Rech, J. A., Nekola, J. C., & Pigati, J. S. (2012). Radiocarbon ages of terrestrial gastropods extend duration of ice-free conditions at the Two Creeks forest bed, Wisconsin, USA. *Quaternary Research*, 77(2), 289–292. https://doi.org/10.1016/j.yqres.2011.11.007
- Richard, P. J., & Occhietti, S. (2005). 14C chronology for ice retreat and inception of Champlain Sea in the St. Lawrence Lowlands, Canada. *Outernary Research*, 63(3), 353–358. https://doi.org/10.1016/j.vgres.2005.02.003
- Ridge, J. C. (1997). Shed brook discontinuity and little falls gravel: Evidence for the Erie interstade in central New York. Geological Society of America Bulletin, 109(6), 652–665. https://doi.org/10.1130/0016-7606(1997)109(0652:SBDALF)2.3.CO;2
- Romé, Y. M., Ivanovic, R. F., Gregoire, L. J., Sherriff-Tadano, S., & Valdes, P. J. (2022). Millennial-scale climate oscillations triggered by deglacial meltwater discharge in last glacial maximum simulations. *Paleoceanography and Paleoclimatology*, 37(10). https://doi. org/10.1029/2022PA004451
- Scargle, J. D. (1982). Studies in astronomical time series analysis. II Statistical aspects of spectral analysis of unevenly spaced data. The Astrophysical Journal, 263, 835. https://doi.org/10.1086/160554
- Schrag, D., Adkins, J., McIntyre, K., Alexander, J. L., Hodell, D. A., Charles, C. D., & McManus, J. F. (2002). The oxygen isotopic composition of seawater during the Last Glacial Maximum. *Quaternary Science Reviews*, 21(1–3), 331–342. https://doi.org/10.1016/S0277-3791(01)00110-X
- Smith, R. S., & Gregory, J. M. (2009). A study of the sensitivity of ocean overturning circulation and climate to freshwater input in different regions of the North Atlantic. *Geophysical Research Letters*, 36(15). https://doi.org/10.1029/2009GL038607

Spratt, R. M., & Lisiecki, L. E. (2016). A Late Pleistocene sea level stack. Climate of the Past, 12(4), 1079–1092. https://doi.org/10.5194/ cp-12-1079-2016

- Stanford, S. D. (2010). Onshore record of Hudson River drainage to the continental shelf from the late Miocene through the late Wisconsinan deglaciation, USA: Synthesis and revision. *Boreas*, 39(1), 1–17. https://doi.org/10.1111/j.1502-3885.2009.00106.x
- Sun, Y., Knorr, G., Zhang, X., Tarasov, L., Barker, S., Werner, M., & Lohmann, G. (2022). Ice sheet decline and rising atmospheric CO₂ control AMOC sensitivity to deglacial meltwater discharge. *Global and Planetary Change*, 210, 103755. https://doi.org/10.1016/j. gloplacha.2022.103755
- Tarasov, L., & Peltier, W. R. (2005). Arctic freshwater forcing of the Younger Dryas cold reversal. *Nature*, 435(7042), 662–665. https://doi.org/10.1038/nature03617
- Taylor, F. B. (1897). Moraines of recession and their significance in glacial theory. *The Journal of Geology*, 5(5), 421-465. https://doi.org/10.1086/607848
- Teller, J. T., & Leverington, D. W. (2004). Glacial lake Agassiz: A 5000 yr history of change and its relationship to the δ18O record of Greenland. Geological Society of America Bulletin, 116(5), 729. https://doi.org/10.1130/B25316.1
- Toney, J. L., Rodbell, D. T., & Miller, N. G. (2003). Sedimentologic and palynologic records of the last deglaciation and Holocene from Ballston Lake, New York. *Quaternary Research*, 60(2), 189–199. https://doi.org/10.1016/S0033-5894(03)00093-0
- Toucanne, S., Soulet, G., Freslon, N., Silva Jacinto, R., Dennielou, B., Zaragosi, S., et al. (2015). Millennial-scale fluctuations of the European Ice Sheet at the end of the last glacial, and their potential impact on global climate. *Quaternary Science Reviews*, *123*, 113–133. https://doi. org/10.1016/j.quascirev.2015.06.010
- Toucanne, S., Zaragosi, S., Bourillet, J. F., Cremer, M., Eynaud, F., Van Vliet-Lanoë, B., et al. (2009). Timing of massive 'Fleuve Manche' discharges over the last 350 kyr: Insights into the European ice-sheet oscillations and the European drainage network from MIS 10 to 2. *Quaternary Science Reviews*, 28(13–14), 1238–1256. https://doi.org/10.1016/j.quascirev.2009.01.006
- Ullman, D. J., Carlson, A. E., Hostetler, S. W., Clark, P. U., Cuzzone, J., Milne, G. A., et al. (2016). Final Laurentide Ice-Sheet deglaciation and Holocene climate-sea level change. *Quaternary Science Reviews*, 152, 49–59. https://doi.org/10.1016/j.quascirev.2016.09.014
- Ullman, D. J., Carlson, A. E., LeGrande, A. N., Anslow, F. S., Moore, A. K., Caffee, M., et al. (2015). Southern Laurentide Ice-Sheet retreat synchronous with rising boreal summer insolation. *Geology*, 43(1), 23–26. https://doi.org/10.1130/G36179.1
- Valdes, P. (2011). Built for stability. Nature Geoscience, 4(7), 414–416. https://doi.org/10.1038/ngeo1200
- VanderPlas, J. T. (2018). Understanding the lomb-scargle periodogram. The Astrophysical Journal Supplement Series, 236(1), 16. https://doi. org/10.3847/1538-4365/aab766
- Vetter, L., Spero, H. J., Eggins, S. M., Williams, C., & Flower, B. P. (2017). Oxygen isotope geochemistry of Laurentide Ice-Sheet meltwater across Termination I. *Quaternary Science Reviews*, 178, 102–117. https://doi.org/10.1016/j.quascirev.2017.10.007
- Wagner, A. J., & Slowey, N. C. (2011). Oxygen isotopes in seawater from the Texas-Louisiana Shelf. Bulletin of Marine Science, 87(1), 1–12. https://doi.org/10.5343/bms.2010.1004
- Wickert, A. D. (2014). Impacts of Pleistocene glaciation and its geophysical effects on North American river systems (Unpublished doctoral dissertation). University of Colorado Boulder.
- Wickert, A. D. (2016). Reconstruction of North American drainage basins and river discharge since the last glacial maximum. Earth Surface Dynamics, 4(4), 831–869. https://doi.org/10.5194/esurf-4-831-2016
- Wickert, A. D. (2019). d18O-ivc-sw: Computing ice-volume-corrected seawater d18O values from sediment-core data. https://doi.org/10.5281/ zenodo.5704598
- Wickert, A. D. (2023). Wickert2023-LaurentideGulfOfMexico: Isotopic evidence for Laurentide Ice Sheet advance and retreat since the last glacial maximum and synthetic paleodrainage history. https://doi.org/10.5281/zenodo.7681780
- Wickert, A. D., Mitrovica, J. X., Williams, C., & Anderson, R. S. (2013). Gradual demise of a thin southern Laurentide Ice Sheet recorded by Mississippi drainage. Nature, 502(7473), 668–671. https://doi.org/10.1038/nature12609
- Williams, C. (2014). A multi-proxy approach to understanding abrupt climate change and Laurentide Ice Sheet melting history based on Gulf of Mexico sediments (Ph.D. Dissertation). University of South Florida.
- Williams, C., Flower, B. P., & Hastings, D. W. (2012). Seasonal Laurentide Ice Sheet melting during the "Mystery Interval" (17.5-14.5 ka). *Geology*, 40(10), 955–958. https://doi.org/10.1130/G33279.1
- Williams, C., Flower, B. P., Hastings, D. W., Guilderson, T. P., Quinn, K. A., & Goddard, E. A. (2010). Deglacial abrupt climate change in the Atlantic warm pool: A Gulf of Mexico perspective. *Paleoceanography*, 115(4), PA4221. https://doi.org/10.1029/2010PA001928
- Willman, H. B., & Frye, J. C. (1970). Pleistocene stratigraphy of Illinois (Vol. 94). Illinois State Geological Survey Bulletin. Retrieved from http://archive.org/details/pleistocenestrat94will
- Zoet, L. K., & Iverson, N. R. (2020). A slip law for glaciers on deformable beds. *Science*, 368(6486), 76–78. https://doi.org/10.1126/science. aaz1183



References From the Supporting Information

- Aharon, P. (2003). Meltwater flooding events in the Gulf of Mexico revisited: Implications for rapid climate changes during the last deglaciation. Paleoceanography, 18(4), 1079. https://doi.org/10.1029/2002PA000840
- Catto, N., Liverman, D. G. E., Bobrowsky, P. T., & Rutter, N. (1996). Laurentide, cordilleran, and montane glaciation in the western peace river — Grande prairie region, Alberta and British Columbia, Canada. *Quaternary International*, 32, 21–32. https://doi. org/10.1016/1040-6182(95)00061-5
- Curry, B. B., Bruegger, A., & Conroy, J. (2018). Highstands and overflow history of glacial Lake Chicago and downstream impacts on Gulf of Mexico δ18O values. *Geology*, 46(7), 667–670. https://doi.org/10.1130/G40170.1
- Curry, B. B., Hajic, E. R., Clark, J. A., Befus, K. M., Carrell, J. E., & Brown, S. E. (2014). The Kankakee Torrent and other large meltwater flooding events during the last deglaciation, Illinois, USA. *Quaternary Science Reviews*, 90, 22–36. https://doi.org/10.1016/j.quascirev.2014.02.006
- Flower, B. P., & Kennett, J. P. (1990). The younger Dryas cool episode in the Gulf of Mexico. *Paleoceanography*, 5(6), 949–961. https://doi. org/10.1029/PA005i006p00949
- Ivanović, R. F., Gregoire, L. J., Wickert, A. D., & Burke, A. (2018). Climatic effect of Antarctic meltwater overwhelmed by concurrent northern hemispheric melt. *Geophysical Research Letters*, 45(11), 5681–5689. https://doi.org/10.1029/2018GL077623
- Kehew, A. E., Teller, J. T., Tellert, J. T., & Teller, J. T. (1994). History of late glacial runoff along the southwestern margin of the Laurentide Ice Sheet. Quaternary Science Reviews, 13(9–10), 859–877. https://doi.org/10.1016/0277-3791(94)90006-X
- Kennett, J. P., Elmstrom, K., & Penrose, N. (1985). The last deglaciation in Orca Basin, Gulf of Mexico: High-resolution planktonic foraminiferal changes. Palaeogeography, Palaeoclimatology, Palaeoecology, 50(2–3), 189–216. https://doi.org/10.1016/0031-0182(85)90068-9
- Knox, J. C. (2007). The Mississippi river system. In A. Gupta (Ed.), Large rivers: Geomorphology and management (pp. 145–182). John Wiley & Sons.
- Lepper, K., Buell, A. W., Fisher, T. G., & Lowell, T. V. (2013). A chronology for glacial Lake Agassiz shorelines along Upham's namesake transect. *Quaternary Research*, 80(1), 88–98. https://doi.org/10.1016/j.yqres.2013.02.002
- Lewis, R. S., & Stone, J. R. (1991). Late quaternary stratigraphy and depositional history of the long island sound basin: Connecticut and New York. *Journal of Coastal Research*, *11*, 1–23. https://doi.org/10.2307/25735570
- Margold, M., Stokes, C. R., Clark, C. D., & Kleman, J. (2015). Ice streams in the Laurentide Ice Sheet: A new mapping inventory. Journal of Maps, 11(3), 380–395. https://doi.org/10.1080/17445647.2014.912036
- Rayburn, J. A., Knuepfer, K., Peter, L., & Franzi, D. A. (2005). A series of large, late Wisconsinan meltwater floods through the Champlain and Hudson Valleys, New York state, USA. *Quaternary Science Reviews*, 24(22), 2410–2419. https://doi.org/10.1016/j.quascirev.2005.02.010
- Rayburn, J. A., Cronin, T. M., Franzi, D. A., Knuepfer, P. L. K., & Willard, D. A. (2011). Timing and duration of North American glacial lake discharges and the Younger Dryas climate reversal. *Quaternary Research*, 75(3), 541–551. https://doi.org/10.1016/j.ygres.2011.02.004
- Richey, J. N., Poore, R. Z., Flower, B. P., & Quinn, T. M. (2007). 1400 yr multiproxy record of climate variability from the northern Gulf of Mexico. Geology, 35(5), 423. https://doi.org/10.1130/G23507A.1
- Rittenour, T. M., Blum, M. D., & Goble, R. J. (2007). Fluvial evolution of the lower Mississippi River valley during the last 100 k.y. glacial cycle: Response to glaciation and sea-level change. *Geological Society of America Bulletin*, 119(5–6), 586–608. https://doi.org/10.1130/B25934.1
- Ross, M., Campbell, J. E., Parent, M., & Adams, R. S. (2009). Palaeo-ice streams and the subglacial landscape mosaic of the North American mid-continental prairies. *Boreas*, 38(3), 421–439. https://doi.org/10.1111/j.1502-3885.2009.00082.x
- Sionneau, T., Bout-Roumazeilles, V., Flower, B. P., Bory, A., Tribovillard, N., Kissel, C., et al. (2010). Provenance of freshwater pulses in the Gulf of Mexico during the last deglaciation. *Quaternary Research*, 74(2), 235–245. https://doi.org/10.1016/j.yares.2010.07.002
- Thirumalai, K., Quinn, T. M., Okumura, Y., Richey, J. N., Partin, J. W., Poore, R. Z., & Moreno-Chamarro, E. (2018). Pronounced centennial-scale Atlantic Ocean climate variability correlated with Western Hemisphere hydroclimate. *Nature Communications*, 9(1), 392. https://doi. org/10.1038/s41467-018-02846-4

Erratum

The originally published version of this article contained an unedited version of Supporting Information S1. The file has been replaced with an edited version, and this may be considered the authoritative version of record.